Author's response to reviews

Title: Using reflexivity to enhance in-depth interviewing skills for the clinician researcher.

Authors:

Ruth P McNair (r.mcnaire@unimelb.edu.au)
Angela Taft (a.taft@latrobe.edu.au)
Kelsey Hegarty (k.hegarty@unimelb.edu.au)

Version: 4 Date: 24 October 2008

Author's response to reviews:

Dear Editor,

I have made the following revisions to the manuscript

1. I have altered the structure the abstract to have the following sections- Background, Results, Conclusions
2. I have removed the text box and replaced it with a table
3. I have removed superscript reference numbers and replaced them with numbers in square brackets.

Yours sincerely

Ruth McNair